Author's response to reviews

Title: Investigation of the Biomechanical Effect of Variable Stiffness Shoe on External Knee Adduction Moment in Various Dynamic Exercises

Authors:

Jee-Chin Teoh (a0040466@nus.edu.sg)
Bena Lim (benalim@nus.edu.sg)
Seung-Bum Park (sbpark@shoenet.org)
Taeyong Lee (bielt@nus.edu.sg)
Jin-Huat Low
Victor P.W. Shim (mpespwv@nus.edu.sg)

Version: 2 Date: 5 June 2013

Author's response to reviews: see over
Dear Professor Hylton Menz

Please find attached our manuscript "Investigation of the Biomechanical Effect of Variable Stiffness Shoe on External Knee Adduction Moment in Various Dynamic Exercises". We would like this study published in the Journal of Foot and Ankle Research.

This study investigates the beneficial effect of variable stiffness shoe on external knee adduction moment in various dynamic exercises.

Also we suggest the following three reviewers who have made significant contribution in the research of foot biomechanics.

Dr Joshua Burns  
Associate Professor  
Institute of Neuroscience and Muscle Research  
The Children’s Hospital at Westmead  
The University of Sydney  
Corner Hawkesbury Rd and Hainsworth St  
Westmead NSW 2145  
Australia  
T +61 2 9845 1228  
email: joshua.burns@health.nsw.gov.au

Dr. Dong Yeon Lee  
Assistant Professor  
Department of Orthopaedic Surgery  
Seoul National University Hospital  
101 Daehak-ro, Jongno-gu
We warrant that the article is original, does not infringe upon any copyright or other proprietary right of any third party, is not under consideration by another journal, and has not been previously published. The authors confirm that we have all reviewed and approved the final version of the manuscript. We look forward to your positive response.

Regards,

Taeyong Lee, Ph.D.
Department of Bioengineering
Block E3A, #07-15A
7 Engineering Drive 1
National University of Singapore
Singapore, 117574
Phone (65) 6516-1471
Fax (65) 6872-3069
E-mail: bielt@nus.edu.sg